\chapter{Introduction}\label{Introduction}

The introduction will be witty and engaging and a joy to read.

I hope.

Points to address:
\begin{enumerate}
	\item What this is.
	\item Why we should care
	\item Most recent development
	\item Do we really mean ``infinite''? (No. Sorry. We lied)
	\item Foundational papers. Historical notes.
\end{enumerate}

See also: The Abstract.
See also: The Conclusion.

The remainder of this report will be organised as follows: Chapters \ref{BackgroundHardware} and \ref{BackgroundSoftware} will discuss approaches to representation of numbers in both Hardware and Software respectively; Chapter \ref{Approach} will outline the approach [reword] considering <insert fuzzy list of faculty dependent requirements here (Safety, Ethics, Group Work Process, etc)>. Chapter/Appendices \ref{Implementation} will provide implementation notes. Chapter \ref{Results} will describe our results. Chapter \ref{Conclusion} will be the conclusion.



NOTE: Given that I don't actually need a full thesis, change from Chapters to Sections?
NOTE2: Will treat as a full thesis but if it gets too big, submit a condensed version as the MCTX "report".

NOTE3: In final version, put the Approach at the end and/or in appendices before the Implementation notes rather than the start (?)
